Author's response to reviews

Title: Primary Placement Technique of Jejunostomy Using the Entristar Skin-Level Gastrostomy Tube in Patients with Esophageal Cancer.

Authors:

Yasushi Rino (rino@med.yokohama-cu.ac.jp)
Norio Yukawa (nryukawa@mac.com)
Hitoshi Murakami (m_air_h@yokohama-cu.ac.jp)
Tsutomu Sato (t-sato@ura.hp.yokohama-cu.ac.jp)
Ken Takata (kemo_t@hotmail.com)
Tsutomu Hayashi (hayashit@kcch.jp)
Takashi Oshima (ohshimatakashi@yahoo.co.jp)
Nobuyuki Wada (wada523@ura.hp.yokohama-cu.ac.jp)
Munetaka Masuda (mmasuda@yokohama-cu.ac.jp)
Toshio Imada (timada@ura.hp.yokohama-cu.ac.jp)

Version: 3 Date: 6 July 2010

Author's response to reviews: see over
I declare that I participated in the design, execution, and analysis of this paper by Yasushi Rino and colleagues entitled "Primary Placement Technique of Jejunostomy Using the Entristar™ Skin-Level Gastrostomy Tube in Patients with Esophageal Cancer." and that I have seen and approved the final version. I also declare that I have no conflict of interest in connection with this paper, other than any noted in the covering letter to the editor.
Dear to the Editor:

I am enclosing herewith our manuscript entitled "Primary Placement Technique of Jejunostomy Using the Entristartm Skin-Level Gastrostomy Tube in Patients with Esophageal Cancer." for publication in your esteemed journal. This study was performed with the approval of an appropriate ethics committee and was carried out according to the Declaration of Helsinki. Informed Consent was obtained from in all patients. The authors declare that we have no competing interests.

I hope that you will see fit to publish this paper in your journal, and we will not hesitate to correct the paper if you have suggestions for its improvement. And if my manuscript is accepted, I would like to use a 20% discount.

Our mailing address is as follows:
Yasushi Rino, M.D.,
Department of Surgery, School of Medicine, Yokohama City University,
Fukuura 3-9, Kanazawa-ku, Yokohama City, 236-0004, Japan

Our telephone number is 81-45-787-2645, Japan, FAX number is 81-45-786-0226, Japan, and E-mail address is rino@med.yokohama-cu.ac.jp.

Yours sincerely,

Yasushi Rino, M.D.,
Department of Surgery,
School of Medicine,
Yokohama City University,
3-9, Fukuura, Kanazawa-ku, Yokohama City, 236-0004, Japan
The email addresses for all the co-authors

Norio Yukawa: nryukawa@mac.com
Hitoshi Murakami: m_air_h@yokohama-cu.ac.jp
Tsutomu Sato: t-sato@urahp.yokohama-cu.ac.jp
Ken Takata: kemo_t@hotmail.com
Tsutomu Hayashi: hayashit@kcch.jp
Takashi Oshima: ohshimatakashi@yahoo.co.jp
Nobuyuki Wada: wada523@urahp.yokohama-cu.ac.jp
Munetaka Masuda: mmasuda@yokohama-cu.ac.jp
Toshio Imada: timada@urahp.yokohama-cu.ac.jp